REVIEW HISTORY

Manuscript Title: From Translation to Naturalization, from Translanguaging to Lingua Universalis

Submitted: 05 February, 2024

Accepted: 10 July, 2024 Published: 31.10.2024

Corresponding Author: Franca Daniele

Email: fdaniele@unich.it

PEER REVIEW

Reviewer 1: Blinded Reviewer 2: Blinded